Author's response to reviews

Title: Self-esteem is associated with premorbid adjustment and positive psychotic symptoms in early psychosis.

Authors:

Kristin Lie Romm (k.l.romm@medisin.uio.no)
Jan Ivar Rossberg (j.i.rossberg@medisin.uio.no)
Charlotte Fredslund Hansen (c.f.hansen@medisin.uio.no)
Elisabeth Haug (Elisabeth.Haug@sykehuset-innlandet.no)
Ole A Andreassen (o.a.andreassen@medisin.uio.no)
Ingrid Melle (ingrid.melle@medisin.uio.no)

Version: 6 Date: 3 August 2011

Author's response to reviews: see over
Dear Production team,

Thank you for your e-mail 2. August 2011 with the proof concerning our manuscript: “Self-esteem is associated with premorbid adjustment and positive psychotic symptoms in early psychosis” (MS: 1225546419548455).

Actually, we do have to make two small changes concerning the 2. Affiliation and the corresponding authors address. Both the University Hospital of Oslo and the University of Oslo has undergone major changes during the last months, and a consequence of this has been several changes of names on departments. In our point-to-point response enclosed in this letter, we have provided the right names for the front page. Hopefully, you will be able to add the correct names now. We are of course, pleased to answer if there should be any questions.

On behalf of the authors,

Kristin Lie Romm
All changes has been marked in the MS with sticky notes:

1) Affiliation #2:

Instead of '2Department of psychiatry, University of Oslo, 0318 Oslo, Norway',

it should read;

'2Institute of Clinical Medicine, Section of Psychiatry, University of Oslo, 0318 Oslo, Norway'

2) Corresponding authors address:

Instead of ‘Section for Psychosis Research.’ it should read;

'Psychosis Research Unit, Division of Mental Health and Addiction, Oslo University Hospital, 0407 Oslo, Norway'